Author's response to reviews

Title: gammaH2AX and Chk1 phosphorylation as predictive pharmacodynamic biomarkers of Chk1 inhibitor-chemotherapy combination treatments

Authors:

Rebecca Rawlinson (rebecca.rawlinson@bath.edu)
Andrew J Massey (a.massey@vernis.com)

Version: 2 Date: 24 April 2014

Author's response to reviews:

Page breaks removed.
Page numbers and line numbers added.
Acknowledgements section added.

Under the methods section, the following has been added: "Ethical Approval. None of the research in this manuscript involved human subjects, human material, or human data, or used regulated vertebrates or invertebrates."